Author's response to reviews

Title: Factors associated with utilisation of GP services by a cohort of people who inject drugs

Authors:

Dhanya Nambiar (dhanya@burnet.edu.au)
Mark Stoove Dr (stoove@burnet.edu.au)
Paul Dietze Professor (pauld@burnet.edu.au)

Version: 2 Date: 5 December 2013

Author's response to reviews: see over
COVER LETTER FOR RESUBMISSION OF MANUSCRIPT

Date: 6th December 2013

Dear Natalie,

Thank you for your feedback regarding the submission of our research manuscript entitled “Factors associated with utilisation of GP services by a cohort of people who inject drugs in Melbourne” for possible publication in the BMC Public Health. Detailed below are changes made to the original document:

1. Addition of names, institutions and email addresses of all authors, and the full postal address of the submitting author
2. Competing interests section included
3. Authors contribution section included
4. Acknowledgments section included

Thank you for your consideration, and I look forward to hearing from you.

Best regards,

Dhanya Nambiar

[Signature]

Professor Paul Dietze
Dr Mark Stoove

Please address correspondence to:

Dhanya Nambiar
PhD Candidate
Burnet Institute
85 Commercial Road, Melbourne, Vic 3004
Email: dhanya@burnet.edu.au